Author’s response to reviews

Title: Lack of knowledge about sexually transmitted infections among women in North rural Vietnam

Authors:

Pham Thi Lan (landhy2003@yahoo.com)
Cecilia Stålsby Lundborg (cecilia.stalsby.lundborg@ki.se)
Ingrid Mogren (ingrid.mogren@obgyn.umu.se)
Ho Dang Phuc (hodang54@yahoo.com)
Nguyen Thi Kim Chuc (ntkchuc@yahoo.com)

Version: 3 Date: 7 April 2009

Author’s response to reviews:

Re: Resubmission of the revised manuscript ID#: 1783610069251642 “Lack of knowledge about sexually transmitted infections among women in North rural Vietnam”

By
Pham Thi Lan, Cecilia Stålsby Lundborg, Ingrid Mogren, Ho Dang Phuc and Nguyen Thi Kim Chuc.

Dear Dr Melissa Norton
Editor-in-Chief
BMC Infectious Diseases,

Thank you for your decision letter dated 13th of March 2009 regarding the above manuscript, and the valuable comments from the reviewers.

In response, we have considered the reviewers’ comments carefully and revised the manuscript accordingly. We are hereby resubmitting the revised manuscript for your consideration. Attached files are: 1) Response to comments from reviewers; 2) The revised manuscript.

The revised version of the manuscript is approved by all co-authors.

We hope that our corrections and adjustments have met your requirements.

We assure that the paper has not been published elsewhere nor submitted to another journal.

We look forward to your reply at your earliest convenience.

Yours sincerely,
Pham Thi Lan
Corresponding author